Author's response to reviews

Title: Single dose of gentamicin in combination with metronidazole versus multiple doses for prevention of post-caesarean infection at Bugando Medical Centre in Mwanza, Tanzania: A Randomized, equivalence, controlled trial

Authors:

Fadhili M Lyimo (fadhili1810@yahoo.com)
Anthony N Massinde (amassinde@gmail.com)
Benson R Kidinya (benkidenya@yahoo.com)
Evelyne T Konje (ekonje28@yahoo.com)
Stephen E Mshana (mshana72@yahoo.com)

Version: 5 Date: 3 May 2013

Author's response to reviews: see over
First Reviewer's report and response from the author

Title: Single dose of gentamicin in combination with metronidazole versus multiple doses for prevention of post-caesarean infection at Bugando Medical Centre in Mwanza, Tanzania: A Randomized, equivalence, controlled trial
Version: 3 Date: 3 April 2013
Reviewer: Baafuor Opoku

Reviewer’s report

There are still a few typographic errors.
1. Line 5 of Page 9--omit the semi-colon after the word 'purulent'
2. Rewrite sentences in lines 11-12 & 13-15 to make them easier to understand.
3. Line 5 of Page 10: all 'data were' instead of all 'data was'

Author’s response

Amended as recommended by reviewer
1. Semicolon added after word PURULENT 2. Sentences line 11-12 & 13-15 re-written 3. it was changed to “Data were”

Second Reviewer's report and author’s response

Reviewer's report

Version: 3 Date: 14 April 2013
Reviewer: Alexander M Aiken
Reviewer's report:
The authors have added one reference (Allegranzi et al) at my suggestion, but the context of this reference seems odd - the reference is a systematic review about HCAI, but it is being used to support the statement that there is a shortage of healthcare-workers in developing countries - these are quite different topics. It might be better to place this reference near the start of the paper to describe the very limited number of studies of this topic in Africa. Otherwise I am satified with the authors responses to my questions

Author’s response

It was good reference, but we couldn’t find the right place to fit it. So we opted to remove it.

Corresponding Author
Dr Anthony Massinde